

Whole Genome Deep Sequencing of HIV-1 Reveals the Impact of Early Minor Variants Upon Immune Recognition During Acute Infection

The Harvard community has made this article openly available. Please share how this access benefits you. Your story matters.

Citation	Henn, Matthew R., Christian L. Boutwell, Patrick Charlebois, Niall J. Lennon, Karen A. Power, Alexander R. Macalalad, Aaron M. Berlin, et al. 2012. Whole genome deep sequencing of HIV-1 reveals the impact of early minor variants upon immune recognition during acute infection. PLoS Pathogens 8(3): e1002529.
Published Version	doi:10.1371/journal.ppat.1002529
Accessed	February 19, 2015 9:56:20 AM EST
Citable Link	http://nrs.harvard.edu/urn-3:HUL.InstRepos:10304558
Terms of Use	This article was downloaded from Harvard University's DASH repository, and is made available under the terms and conditions applicable to Other Posted Material, as set forth at http://nrs.harvard.edu/urn-3:HUL.InstRepos:dash.current.terms-of- use#LAA

(Article begins on next page)

Whole Genome Deep Sequencing of HIV-1 Reveals the Impact of Early Minor Variants Upon Immune Recognition During Acute Infection

Matthew R. Henn¹, Christian L. Boutwell²⁹, Patrick Charlebois¹⁹, Niall J. Lennon¹, Karen A. Power², Alexander R. Macalalad¹, Aaron M. Berlin¹, Christine M. Malboeuf¹, Elizabeth M. Ryan¹, Sante Gnerre¹, Michael C. Zody¹, Rachel L. Erlich¹, Lisa M. Green¹, Andrew Berical², Yaoyu Wang², Monica Casali¹, Hendrik Streeck², Allyson K. Bloom², Tim Dudek², Damien Tully², Ruchi Newman¹, Karen L. Axten², Adrianne D. Gladden², Laura Battis², Michael Kemper², Qiandong Zeng¹, Terrance P. Shea¹, Sharvari Gujja¹, Carmen Zedlack³, Olivier Gasser⁴, Christian Brander^{5,6}, Christoph Hess⁴, Huldrych F. Günthard⁷, Zabrina L. Brumme², Chanson J. Brumme², Suzane Bazner⁸, Jenna Rychert⁸, Jake P. Tinsley⁹, Ken H. Mayer⁹, Eric Rosenberg⁸, Florencia Pereyra², Joshua Z. Levin¹, Sarah K. Young¹, Heiko Jessen³, Marcus Altfeld², Bruce W. Birren¹, Bruce D. Walker^{2,10}, Todd M. Allen^{2*}

1 Broad Institute of MIT and Harvard, Cambridge, Massachusetts, United States of America, 2 Ragon Institute of MGH, MIT and Harvard, Boston, Massachusetts, United States of America, 3 HIV Clinic Praxis. Jessen, Berlin, Germany, 4 Immunobiology Lab, Department of Biomedicine, University Hospital Basel, Basel, Switzerland, 5 Institució Catalana de Recerca i Estudis Avançats (ICREA), Barcelona, Spain, 6 Irsicaixa AIDS Research Institute-HIVACAT, Hospital University Germans Trias I Pujol, Badalona, Spain, 7 Division of Infectious Diseases and Hospital Epidemiology, University Hospital Zurich, University of Zurich, Switzerland, 8 Massachusetts, United States of America, 9 The Fenway Institute, Fenway Health, Boston, Massachusetts, United States of America, 10 Howard Hughes Medical Institute, Chevy Chase, Maryland, United States of America

Abstract

Deep sequencing technologies have the potential to transform the study of highly variable viral pathogens by providing a rapid and cost-effective approach to sensitively characterize rapidly evolving viral guasispecies. Here, we report on a highthroughput whole HIV-1 genome deep sequencing platform that combines 454 pyrosequencing with novel assembly and variant detection algorithms. In one subject we combined these genetic data with detailed immunological analyses to comprehensively evaluate viral evolution and immune escape during the acute phase of HIV-1 infection. The majority of early, low frequency mutations represented viral adaptation to host CD8+ T cell responses, evidence of strong immune selection pressure occurring during the early decline from peak viremia. CD8+ T cell responses capable of recognizing these low frequency escape variants coincided with the selection and evolution of more effective secondary HLA-anchor escape mutations. Frequent, and in some cases rapid, reversion of transmitted mutations was also observed across the viral genome. When located within restricted CD8 epitopes these low frequency reverting mutations were sufficient to prime de novo responses to these epitopes, again illustrating the capacity of the immune response to recognize and respond to low frequency variants. More importantly, rapid viral escape from the most immunodominant CD8+ T cell responses coincided with plateauing of the initial viral load decline in this subject, suggestive of a potential link between maintenance of effective, dominant CD8 responses and the degree of early viremia reduction. We conclude that the early control of HIV-1 replication by immunodominant CD8+ T cell responses may be substantially influenced by rapid, low frequency viral adaptations not detected by conventional sequencing approaches, which warrants further investigation. These data support the critical need for vaccine-induced CD8+ T cell responses to target more highly constrained regions of the virus in order to ensure the maintenance of immunodominant CD8 responses and the sustained decline of early viremia.

Citation: Henn MR, Boutwell CL, Charlebois P, Lennon NJ, Power KA, et al. (2012) Whole Genome Deep Sequencing of HIV-1 Reveals the Impact of Early Minor Variants Upon Immune Recognition During Acute Infection. PLoS Pathog 8(3): e1002529. doi:10.1371/journal.ppat.1002529

Editor: Christopher M. Walker, Nationwide Children's Hospital, United States of America

Received October 2, 2011; Accepted December 27, 2011; Published March 8, 2012

Copyright: © 2012 Henn et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This project was funded in part with Federal funds from the Bill and Melinda Gates Foundation (BDW, TMA), the National Institute of Allergy and Infectious Disease, the National Institutes of Health, and the Department of Health and Human Services under grants NIAID P01-Al074415 (TMA, MA), NIAID T32-Al07245 (CLB), and contracts no. HHSN272200900018C (BWB) and HHSN272200900006C (BWB), as well as by an Innovation Award from the Ragon Institute of MGH, MIT and Harvard (TMA and MRH) and by the Swiss HIV Cohort Study (SHCS) project No. 535 (OG, CH, and HFG). We also thank the International HIV Controllers Study and the SHCS, supported by the Swiss National Science Foundation (SNF # 33CSC0-108787) for collaboration.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: tallen2@partners.org

• These authors contributed equally to this work.

Author Summary

The ability of HIV-1 and other highly variable pathogens to rapidly mutate to escape vaccine-induced immune responses represents a major hurdle to the development of effective vaccines to these highly persistent pathogens. Application of next-generation or deep sequencing technologies to the study of host pathogens could significantly improve our understanding of the mechanisms by which these pathogens subvert host immunity, and aid in the development of novel vaccines and therapeutics. Here, we developed a 454 deep sequencing approach to enable the sensitive detection of lowfrequency viral variants across the entire HIV-1 genome. When applied to the acute phase of HIV-1 infection we observed that the majority of early, low frequency mutations represented viral adaptations to host cellular immune responses, evidence of strong host immunity developing during the early decline of peak viral load. Rapid viral escape from the most dominant immune responses however correlated with loss of this initial viral control, suggestive of the importance of mounting immune responses against more conserved regions of the virus. These data provide a greater understanding of the early evolutionary events subverting the ability of host immune responses to control early HIV-1 replication, vielding important insight into the design of more effective vaccine strategies.

Introduction

A major challenge to the development of effective vaccines against highly variable viruses is their ability to adapt to evade host immune responses [1–4]. During HIV-1 infection, for example, immune escape mutations develop which impair the ability of both CD8+ T cell responses and neutralizing antibodies to maintain immune control [5–9]. However, some CD8+ T cell escape mutations have been shown to dramatically impair viral replication capacity, which may slow viral escape and contribute significantly to the ability of some responses to effectively control HIV-1 [10–13]. The outcome of this dynamic interplay between immune responses functioning to eliminate infected cells, emerging escape variants that evade these responses, and the impact of these variants on viral replication, critically influences early immune control of HIV-1.

The majority of studies on HIV-1 evolution have relied on bulk Sanger sequencing to define the major genetic variants that arise during infection. These studies have demonstrated that upwards of 50% of mutations observed over the course of infection may be associated with viral adaptations to CD8+ T cell responses [5,14]. Unfortunately, bulk Sanger sequencing is insufficient to detect low frequency variants that are particularly important during the acute phase of infection when viral escape occurs rapidly. The application of single genome amplification and sequencing (SGA or SGS) has increased the sensitivity for detecting and quantifying low frequency viral variants [7,15,16] but high cost and poor scalability limit its broader application. As a result, a sensitive and comprehensive understanding of the genetic pathways and kinetics of viral adaptation to acute phase immune selection pressures across the entire HIV-1 genome, likely a critical determinant of the success or failure of both natural and vaccine-elicited immune responses, is lacking.

Next-generation sequencing (NGS) or deep sequencing approaches such as 454 pyrosequencing [17] have the potential to transform the study of HIV-1 and other highly variable pathogens

by providing a rapid and cost-effective approach for the sensitive characterization of the complex and rapidly evolving intra-patient viral quasispecies. Recent studies have applied deep sequencing approaches to HIV-1 and SIV to detect low frequency drug resistance variants [18-22] and CD8+ T cell escape variants [23-26], although these studies have largely been limited to the analysis of discrete regions of interest. Here we report an approach for routine whole genome sequencing of HIV-1 that combines deep sequencing with novel algorithms for *de novo* sequence assembly and for accurate quantification of low frequency viral variants. This new platform not only provides the capacity to rapidly sequence across the whole HIV-1 genome for population-scale genetic analyses of large cohorts of HIV-1 infected individuals, but also the sensitivity to comprehensively characterize the earliest stages of viral immune adaptation during the critical initial interactions with the host immune response. The application of this whole genome deep sequencing platform to longitudinal samples from a single subject during acute HIV-1 infection reveals the speed and complexity of the simultaneous adaptation of HIV-1 to multiple host immune responses, and suggests that early, low frequency escape variants to dominant acute-phase CD8+ T cell responses may have a significant impact on the early immune control of HIV-1.

Results

Whole HIV-1 genome sequencing and assembly using 454 pyrosequencing

Prior studies utilizing deep sequencing to more critically examine HIV-1 and SIV sequence diversity and evolution have focused predominantly on short, specific regions of the virus where evolution was known or expected to occur. To apply a deep sequencing approach that can interrogate diversity across the complete genome we designed primers that amplify four overlapping PCR amplicons spanning the entire protein-coding region of the HIV-1 genome (HXB2 nt 779-9551; Figure 1A) and validated them against a set of 89 HIV-1 clade B (HIV-1B) clinical samples from subjects in the acute and chronic phase of infection, as well as low-viremia controllers (Table S1 in Text **S1**). To reduce costs, we pooled the four amplicons from each individual sample prior to acoustic shearing and subject-specific molecular bar-coding, and then batched bar-coded samples from multiple subjects prior to performing emulsion PCR and pyrosequencing.

In contrast to traditional Sanger sequence data, the 454 sequence data provides deep read coverage (sequencing reads per site) where each individual base and the context in which it occurs in the read can be leveraged to inform the consensus assembly. As such, we developed AssembleViral454 (AV454), a module in the ARACHNE^{17,18} assembly tool kit (see Supplementary Methods in Text S1), which takes advantage of deep sequence coverage and the knowledge that continuous RNA viral genomes do not generally contain large repetitive sequences to correctly assemble all reads. As shown in Figure 1B, AV454 consensus assemblies captured on average $96.3\% \pm 11.3\%$ (s.d.; n = 89) of patient-specific reads into a single contig (**Table S1** in Text S1), significantly outperforming the other assemblers. While both AV454 and *Newbler* captured >98% of the target genome by all contigs assembled, AV454 captured a significantly greater percentage of the genome in a single continuous contig than any other assembler (see Supplementary Methods in Text S1; Wilcoxon, p < 0.001, n = 67) and exhibited a much tighter distribution of results. These data demonstrate the ability of this sequencing and assembly strategy to reproducibly generate



Figure 1. PCR amplification strategy and performance of novel assembly, read alignment, and variant detection algorithms. (A) PCR amplification strategy using four ~3.2 kb amplicons spanning gag through nef of the HIV-1 genome. Amplicons were then pooled, sheared, barcoded by patient or time point, and batched for library construction and single-molecule 454 pyrosequencing. (B) AssembleViral454 v1.0 outperforms other algorithms in its ability to assemble de novo continuous consensus contigs that span the complete target region. Results are shown for 67 acute, chronic, or controller patient samples that had successful amplification of all four amplicons and at least 10-fold sequence coverage (sequencing reads per site) across >70% of the target genome. Black lines denote the mean score for each assembler, red line the median, red box ends the 25th and 75th quantiles, and red box whiskers the upper and lower quartiles plus/minus 1.5 times the interquartile range, respectively. (C) ReadClean454 v1.0 corrects for read alignment errors due to various sequence error modes and significantly reduces process error rate. Results shown are for virus from two infectious clones, NL43 (WT) and NL43 (RKLM) containing two point mutations in Gag [10], sequenced independently to 417- and 189-fold average coverage, respectively. Errors are defined as base calls or InDels that differ from the assembled consensus at a given position, and the read error rate is the total number of errors per total number of NQS passing bases interrogated. Percentage of reads on which a correction was made at each step are shown in *parentheses*. A final average process error rate of 0.5×10^{-4} was achieved based on both infectious clones. (D) V-Phaser v1.0, utilizes phasing information to identify a variant pair found in 1.0% of the reads covering both loci when there are 200 such reads; without phase, a three-fold increase in coverage is required to achieve the same 1.0% detection threshold. A variant at a frequency of 0.1% can be detected when phased coverage is 2999-fold. doi:10.1371/journal.ppat.1002529.g001

genome-wide sequence assemblies from a wide variety of different HIV-1B clinical isolates.

Development of novel algorithms to accurately detect low frequency variants

A major challenge to the utility of deep sequence data is distinguishing true genetic polymorphisms from process errors [22,23,26,27]. We addressed this problem by developing an analysis pipeline designed to: (i) maximize the read data retained for analysis, (ii) optimize read alignments, and (iii) leverage phase information to improve the sensitivity and specificity of variant calling. First, all read alignments are made to the sample's AV454consensus assembly. A comparison of read alignments to the AV454 de novo assembly versus an HIV-1B reference sequence demonstrated that use of the AV454 assembly retained more reads and bases for analysis and significantly reduced the number of insertions and deletions that result in alignments with frame shifts (Table S2 in Text S1; Wilcoxon, p<0.0001), an important consideration for variant calling. Second, ReadClean454 (RC454) applies a Neighborhood Quality Standard (NQS) base filter [28], corrects reads for common process errors such as homopolymer and carry-forward-incomplete-extension (CAFIE) miscalls (see Supplementary Methods and Figure S1 in Text S1), and further optimizes read alignments using coding frame information. As shown in Figure 1C, RC454 significantly reduces the average process read error rate from 1.3×10^{-2} to 0.5×10^{-4} errors per base as determined by the sequencing of infectious HIV-1 clones. Next, V-Phaser distinguishes true variants from sequencing errors by defining the frequency at which a nucleotide polymorphism must be observed to be considered a true variant. This is accomplished through the application of an error probability model initially defined by a uniform empirical process read error

rate and then refined by the inclusion of variant nucleotide phasing information i.e. correlated sequence changes (see Supplementary Methods in Text S1; Macalalad et al, manuscript submitted). Lastly, V-Profiler calculates the frequency of each triplet codon composed of nucleotides accepted by V-Phaser. When applied to samples of known composition, this pipeline quantified variants with high sensitivity (100%) and specificity (97%), and implementation of the phasing-based approach achieved detection of 1.0% variants when \geq 200-fold shared sequence coverage (sequencing reads per site) was attained; this represents a three-fold decrease in required coverage over non-phase based methods; Figure 1D; see Supplementary Methods in Text S1). The application of these algorithms provides the ability to rapidly characterize intra-patient HIV-1 genetic diversity, and facilitate the routine handling of deep sequencing data for whole genome assembly and variant detection, as shown in Figure S2 in Text S1 for all HIV-1 proteins from the array of 89 HIV-1B clinical samples.

Benchmarking of 454 sequencing to traditional cloning and SGA

The whole HIV-1 genome 454 deep sequence platform was validated by comparison to bulk Sanger sequencing, cloning and sequencing, and SGA. First, we compared full length consensus HIV-1 sequences for four longitudinal samples from a single subject (9213) generated by bulk Sanger sequencing and by the 454 platform (35,093 total nucleotides compared; see Supplementary Results in **Text S1**). Overall, the Sanger and 454 consensus sequences differed at only six nucleotides and one insertion/ deletion (InDel), and in each case the discrepancy resulted from a differential consensus call at a highly polymorphic position (Table S3A and S3B in Text S1). Next, we extensively compared variant quantification across a highly variable 1544 nucleotide region spanning from vif to tat in a single sample (subject 9213) by deep sequencing (average 566-fold high quality sequencing reads per site), traditional PCR cloning and sequencing (768 clones), and single genome amplification (87 single genomes). We observed 95.6% concordance between the three methods in the detection of invariant/variant sites (see Supplementary Results in Text S1), and the calculated variant frequencies were highly correlated between methods as shown for deep sequencing vs cloning and sequencing in Figure 2. These data confirm the ability of this



Figure 2. Comparison of sequence variant quantification by 454 deep sequencing and by PCR cloning/sequencing. Orthogonal regression of variant frequency estimates obtained by 454 and clonal sequence data across the highly variable 1544 nucleotide region spanning Vif to Tat in subject 9213 (slope = 1.01; 95% Cl, 0.73 to 1.40). doi:10.1371/journal.ppat.1002529.g002

high-throughput, deep sequencing platform to profile HIV-1 quasispecies diversity as accurately as conventional cloning and sequencing or SGA.

Characterization of whole HIV-1 genome evolution during acute infection

Recent studies utilizing deep sequencing to more sensitively assess early, low frequency variants within specific CD8 epitopes reveal that viral escape from CD8+ T cell responses can occur very rapidly [23,25,26], even as soon as 17 days following SIV infection of macaques [23]. To further explore the dynamics of HIV-1 evolution and immune adaptation during acute infection, we conducted a comprehensive and sensitive assessment of early viral evolution, without bias towards previously studied epitopes, by producing longitudinal genome-wide 454 sequence data from longitudinal samples from a single subject identified as HIV-1 infected prior seroconversion. Subject 9213 presented with a baseline viral load of 9.3 million copies/ml (day 0 postpresentation) that peaked at 21 million copies/ml on day 3 (Figure 3A). A negative Western blot on day 0 supported likely infection within 15–20 days of first sampling, i.e. Fiebig stage II-III [29]. We captured genetic diversity data for the entire open



Figure 3. Clinical course and whole genome deep sequence coverage for subject 9213. (A) Clinical course of infection in subject 9213 shown as days post-presentation. Plasma viral load (copies/ml) is shown in *red* and CD4+ T cell count (cells/ml) in *blue*. Estimated acute/early Fiebig stages are shown and *arrows* indicate time points sequencing reads per site across the HIV-1 genome for subject 9213 at six time points (days post presentation). Reads are aligned to the consensus assembly of their respective time point using *Mosaik v1.0* (Table S9 in Text S1) and coverage (sequencing reads per site) calculated from bases that pass the defined Neighbor Quality Standard (NQS, see Supplementary Methods in Text S1) [28]. PCR amplicon locations are denoted by horizontal bars under the x-axis. doi:10.1371/journal.ppat.1002529.g003

reading frame of HIV-1 at six time points over the first 4 years of infection (day 0, 3, 59, 165, 476, 1543) at an average number of sequencing reads per site of 535±325 reads (Figure 3B, Table S1 in Text S1). Codon diversity, defined as the frequency of codons that differed from the consensus codon at baseline (day 0), was calculated for each position of the HIV-1 proteome. As illustrated in Figure 4A and 4B, there was strikingly little codon diversity present in the viral population during peak viremia, with less than 2% and 5% of all positions exhibiting detectable diversity at day 0 and day 3, respectively, and of those positions that did vary the majority varied by less than 2%. The low genetic diversity of the viral quasispecies during early acute infection, which would not have been discernable using traditional bulk sequencing approaches, confirms that infection in this subject was founded by a single genetic lineage, in line with recent reports suggesting that most sexually transmitted HIV-1 infections arise from a single founder virus [15,16,30-33].

Early HIV-1 evolution is associated with immune adaptation

The first evidence of HIV-1 evolution was observed at day 59, when 11% of all codons exhibited detectable diversity

(Figure 4C). However, still only a minor subset of 21 codons exhibited any substantial (>10%) degree of variation from baseline at this time point when peak viral loads were observed to dramatically decline to 298,000 copies/ml (Figure 3A). Although as expected the number of evolving codons continued to increase over time, with 38 exhibiting detectable diversity at day 165 (Figure 4D), it is notable that over half of the day 59 sites exhibiting substantial variation (>10%) declined in variation by day 165 (Figure 5). These data reveal complexities in the early evolution of the viral quasispecies that are not typically observed by traditional sequencing methods. Moreover, as shown in Figure 5 even by day 165 no single codon had yet mutated towards fixation (>95%), suggesting that the substantial early decline in peak viremia in subject 9213 was not associated with any dramatic turnover of the viral population.

Given that CD8+ T cell responses represent a major driving force of viral evolution following acute HIV-1 infection [5,7,34], we examined the extent to which these early, low frequency mutations might represent viral adaptation to cellular immune responses. Here we compared amino acid divergence from baseline within described CD8+ T cell epitopes restricted by subject 9213's HLA alleles to the amino acid divergence at all



Figure 4. Rapidly expanding sequence diversity during HIV-1 infection. Heat maps illustrate sites exhibiting amino acid sequence diversity at days 0, 3, 59, 165, 476 and 1543 post-presentation. Plotted is the percentage of amino acid diversity at each position with respect to the dominant baseline (day 0) amino acid residue. All 3174 amino acids of HIV-1 are represented, with the first amino acid of Gag located in the top left corner of the grid and the last amino acid of Nef located in the bottom right corner. Completely conserved residues are *dark blue*, low-level variant residues (<10% divergent from baseline) are *light blue*, moderately variable residues (10–50%) in *orange*, and highly variant residues (>50%) in *red*. (**A**) 0 days p.p., (**B**) 3 days p.p., (**C**) 59 days p.p., (**D**) 165 days p.p., (**E**) 476 days p.p., (**F**) 1543 days p.p..



Figure 5. Limited evolution in the HIV-1 proteome prior to establishment of viral set point. Sequence diversity is plotted for all evolving codons in each HIV-1 protein as the percent of sequences with an amino acid residue different from the dominant baseline residue. Colored lines denote individual evolving amino acid residues within each protein. The time of infection prior to the establishment of viral set point (day 165) is highlighted in *grey.* doi:10.1371/journal.ppat.1002529.q005

other positions across the proteome. We observed that the majority of early viral evolution at days 59 and 165 was indeed shaped by cellular immune responses, with significantly greater diversity observed within restricted epitopes (Wilcoxon, p = 0.016; **Figure 6A**; **Table S4** in **Text S1**). At day 59, this was most pronounced in Vif and Nef, with Env and Pol also exhibiting diversity preferentially within restricted CD8 epitopes by day 165. These data suggest that rapid adaptation to cellular immune responses was the major driving force for the early, low frequency viral evolution observed in subject 9213.

Rapid viral escape from immunodominant acute phase CD8+ T cell responses

To better understand the early immune adaptation of HIV-1 in subject 9213, we characterized the breadth and magnitude of CD8+ T cell responses to all 19 described CD8+ T cell epitopes by IFN-gamma ELISPOT assay using autologous peptides. Acute phase (day 59) responses were detected against six epitopes, with the two most dominant responses directed against the Vif B38-WI9 (2744 SFC/Mill PBMC) and Nef A24-RW8 (2584 SFC) epitopes, while weaker subdominant responses were directed against the Pol B44-EW9 (814 SFC), Rev A01-IY9 (734 SFC), Gag B44-AW11 (444 SFC), and Gag A01-GY9 (144 SFC) epitopes (**Table S5** in **Text S1**). The deep sequencing data revealed evidence of viral adaptation, i.e., escape, within four of the six epitopes (**Figure 7 and Table S6** in **Text S1**). The escape phenotype of the observed genetic variants was confirmed by the impaired recognition of each of the variant's peptides when tested in IFN-gamma ELISPOT assays (**Table S5** in **Text S1**).

Viral immune adaptation was most rapid in the dominantly targeted Vif B38-WI9 and Nef A24-RW8 epitopes, with estimated escape rates of 0.0987 day⁻¹ and 0.0976 day⁻¹, respectively (Figure 6B). Interestingly, we observed distinct adaptive pathways by which the virus evaded each of these dominant, early responses. In the Vif B38-WI9 epitope, by day 59 56.6% of the viral population expressed one of four intra-epitope mutations (Figure 6C), and/or three flanking mutations likely affecting antigen processing [35,36]. This initial apparent exploration of multiple escape pathways resolved over time with over 98% of sequenced reads from the population now comprising just three variant "haplotypes" by day 165, before fixing on the I87V mutation at the C-terminal HLA-class I epitope anchor residue by day 476 (Figure 7). In contrast, immune adaptation in the Nef A24-RW8 epitope followed a more restricted pathway with 54.7% of the quasispecies expressing a single escape mutation $(F_{148}L)$ at day 59, followed by the emergence of a second escape mutation $(T_{147}M)$ at day 165 which together comprised >99% of the total population (**Figure 6D**). Interestingly, the original $F_{148}L$ mutation at position 6 of the epitope was replaced by day 476 with the $Y_{144}F$ mutation, a position 2 HLA-anchor mutation that is likely a more potent escape mutation. This approximately 50-50 mixed population of position 2 $Y_{144}F$ and position 5 $T_{147}M$ escape variants remained stable out to day 1543 (Figure 7). Thus, deep sequencing during the acute phase of infection revealed rapid viral



Figure 6. Cellular immune responses drive early low-frequency quasispecies diversity. (**A**) For each protein, the average frequency of non-dominant baseline residues of positions within the 19 described CD8 epitopes restricted by subject 9213's HLA alleles (*left*) and outside of the 19 described epitopes (*right*) is plotted for each time point sequenced. *Colored lines* denote the proteins for which diversity was substantially higher inside of CD8 epitopes versus outside CD8 epitopes. (**B**) To determine rates of viral escape for each epitope escape mutations were defined as any amino acid substitution within the epitope. *Symbols* denote the cumulative observed frequency of all escape mutations, and lines depict the best fit by non-linear regression of the observed frequency data to the CTL escape model of Asquith et al. [65]. *Open symbols* and dashed lines denote epitopes for which evolution was consistent with reversion. *Black symbols* and *dotted lines* denote epitopes for which there was no evidence of escape. CD8 responses against each epitope are shown in parentheses in the legend and were measured by IFN-gamma Elispot assay (Spot Forming Cells/Mill PBMC (SFC)). (**C**) Frequency of wild-type (*black*) and variant (*red*) haplotypes of the Vif B38-WI9 epitope and flanking regions over time. *Blue* residues highlight differences between the day 0 transmitted sequence and HIV-1B consensus sequence.

doi:10.1371/journal.ppat.1002529.g006



Figure 7. Viral escape from acute and chronic phase CD8+ T cell responses. Stacked heat-maps illustrate variant codon frequencies over time for each residue of the CD8 epitopes targeted by subject 9213. Shown are epitopes targeted during the acute (Day 59) and chronic (Day 476) phases of HIV-1 infection. The baseline sequence is shown at the top of each epitope, with non-HIV-1B consensus residues highlighted in *blue.* The magnitude of each response is shown in SFC per million PBMC. doi:10.1371/journal.ppat.1002529.g007

escape from the two most dominant acute phase CD8+ T cell responses, in some cases through the combined effects of multiple low frequency variants that would be missed by traditional bulk sanger sequencing. Interestingly, in both cases the early escape mutations were ultimately replaced in the viral population by HLA anchor position mutations that more efficiently escaped immune recognition (**Table S5** in **Text S1**), presumably through reductions in MHC-I:peptide binding at the cell surface.

Slower rate of escape from subdominant CD8+ T cell responses

Viral escape was also observed in the Pol B44-EW9 and Gag A01-GY9 epitopes (**Figure 7 and Table S6** in **Text S1**) that were targeted by subdominant acute-phase CD8+ T cell responses of 814 SFC and 144 SFC, respectively (**Table S5** in **Text S1**). Here, the lower magnitude of these responses was associated with slower estimated escape rates of 0.0133 and 0.0036 day⁻¹ (**Figure 6B**). In both cases, early, low frequency mutations at

positions 4 and 6 of these epitopes at day 165, likely T cell receptor (TCR) escape mutations, were subsequently replaced by HLAanchor mutations at position 2 or 9. These data provide insight into possible mechanisms underlying the transient variation of some residues observed in **Figure 5**, whereby early mutations are being out competed by more effective secondary mutations. Finally, the two other epitopes that were targeted during acute infection, Rev A01-IY9 (734 SFC) and Gag B44-AW11 (444 SFC), exhibited no evidence of immune escape over the course of infection despite the higher sensitivity of deep sequencing (**Figure 7 and Table S6** in **Text S1**).

In addition to the six epitopes targeted during acute infection, weak CD8+ T cell responses were also detected against four other epitopes during the chronic phase of infection (day 476): Env Cw4-SF9 (190 SFC), Env A24-RL9 (84 SFC), Env A01-RY9 (80 SFC), and Nef A01-YT9 (70 SFC) (**Table S5** in **Text S1**). There was evidence of viral escape in the three Env epitopes (**Figure 7 and Table S6** in **Text S1**), but similar to the epitopes targeted by subdominant acute phase responses, the rate of escape in these chronically targeted epitopes was slow at 0.0067, 0.0087, and 0.0026 day⁻¹ respectively (**Figure 6B**). Overall, the virus escaped from four of the six CD8+ T cell responses mounted during the acute phase of infection and three of the four CD8+ T cell responses mounted during chronic infection, with highly variable rates of escape observed for different epitopes.

Correlation between CD8+ T cell immunodominance and rate of escape

In subject 9213, we found that the rate of immune escape from acute phase CD8+ T cell responses correlated with the magnitude of these responses (p = 0.01), reflecting the differential selective pressure imposed on the viral population by distinct CD8+ T cell responses. Interestingly, greater than 50% of the viral population had escaped the dominantly targeted Vif B38-WI9 and Nef A24-RW8 epitopes by 59 days post-presentation, which corresponds temporally to the plateauing of the precipitous decline from peak viremia and the subsequent equilibration of viral load (**Figure 3A**). Thus, these data from a single subject suggest that the rate at which the virus escapes from critical acute phase immunodominant responses, in some cases through the combined effects of multiple low frequency mutations, may influence the magnitude of the drop from peak viremia and duration of effective early immune control, and by extension set-point viral load.

Low frequency escape mutations are associated with variant-specific CD8+ T cell responses

We have previously observed that CD8+ T cell responses can arise that are capable of recognizing CTL escape variants [37–39], demonstrating that the immune system is at least partially able to contend with immune escape. To investigate the kinetics of such variant-specific responses, and whether they might be triggered by early, low frequency mutations arising during the acute phase of infection, we screened for responses against the most frequent escape variants in the rapidly escaping Vif B38-WI9 and Nef A24-RW8 epitopes. As early as day 59, strong responses were detected against two of the primary escape variants in the Vif B38-WI9 epitope, despite the fact that the S₈₆A and I₈₇V mutations comprised less than 15% of the viral quasispecies (**Figure 8A**; **Table S5** in **Text S1**). These variant-specific responses persisted out to day 476, and as we have previously observed in chronic infection were equal in magnitude to the autologous wild-type response [38]. However, fixation in the epitope of the C-terminal I₈₇V mutation, likely impairing MHC-I binding and presentation, ultimately coincided with a significant (>10-fold) decline of both wild-type and the variant-specific responses by day 661. We also detected early responses to escape variants in the Nef A24-RW8 epitope, albeit at much lower magnitudes, and similarly the emergence of an HLA anchor position escape mutation $(Y_{144}F)$ ultimately abrogated responses against both wild-type and variant peptides (Figure 8B; Table S5 in Text S1). This early recognition of the low-frequency escape variants, followed by loss of responses upon outgrowth of HLA anchor mutations, suggests partial cross-recognition of early escape variants by the wild-typespecific response [40] rather than development of de novo variantspecific responses [37]. Thus, eventual loss of the wild-type sequence, required for continuous expansion of these wild-typespecific responses, results in the eventual decline of all responses. Thus, these data extend earlier reports of the ability of CD8+ T cell responses to recognize viral escape mutations [37-40] by illustrating the ability of early responses to recognize low frequency escape mutations and providing a mechanism for the observed substitution of early escape mutations with more potent secondary HLA-anchor mutations.

Reversion of transmitted escape mutations primes CD8+ T cell responses

Apart from the evolution in the targeted CD8 epitopes described above, we also observed substantial evolution in four other non-targeted CD8 epitopes restricted by subject 9213's HLA alleles (Nef A01-WH10, Env B44-AY10, Gag A24-KW9, and Gag B44-LY9; Table S7 in Text S1). Responses were never detected against these epitopes during either acute (day 59) or chronic (day 476 and day 661) infection despite testing with autologous peptides matching the founder virus (Table S5 in Text S1). Each of these evolving epitopes was found to contain one or more transmitted mutations at baseline (day 0), with the observed evolution consistent with the reversion of these transmitted mutations back towards the HIV-1B consensus sequence. Reversions in the Nef A01-WH10 and Env B44-AY10 epitopes occurred with estimated rates of 0.0722 and 0.0887 day^{-1} (Figure 9A), respectively, nearly equaling those of the most rapidly escaping Vif B38-WI9 and Nef A24-RW8 epitopes (Figure 6B). Reversion in the Gag A24-KW9 epitope was actually the result of the transmission and reversion of



Figure 8. Variant-specific CD8+ T cell Elispot responses. Elispot responses in Spot Forming Cells (SFC) per million PBMC to wild-type and variant peptides for the two dominant epitopes (A) Vif B38-WI9 (WHLGQGVSI) and (B) Nef A24-RW8 (RYPLTFGW). Bars in *black* denote responses to clade B consensus epitopes. Bars in *red, orange, and pink* denote responses to epitopes containing escape variants. doi:10.1371/journal.ppat.1002529.g008



Figure 9. Reversion of transmitted mutations over the course of infection. (**A**) Rates of reversion of transmitted mutations within both restricted and unrestricted CD8 epitopes in subject 9213. Reversion was defined as the replacement of a transmitted non-consensus residue by the HIV-1B consensus residue. *Symbols* denote the observed frequency of viruses expressing the consensus residue and *lines* depict the best fit by non-linear regression of the observed frequency data to the CTL escape model of Asquith et al. [65]. *Closed symbols* and *solid lines* denote epitopes restricted by subject 9213, while *open symbols* and *dashed lines* denote epitopes not restricted by 9213. Listed in *parentheses* are the mutations listed by the consensus residue, HXB2 position, followed by the transmitted mutation, i.e., F1975. (**B**) Rates of reversion of all transmitted mutations exhibiting sequence variation over the course of infection. Each line represents a different mutation. doi:10.1371/journal.ppat.1002529.g009

a K₂₈Q mutation that is a well-described escape mutation in the overlapping A03-RK9 epitope [35,41]. Interestingly, founder virus mutations reverted in three additional HLA-A03 epitopes (Pol-ATK9, Pol-QK9, and Vif-RK10), and two HLA-B57 epitopes (Vif B57-IF9 and Nef B57-YY9), suggesting that the founder virus in subject 9213 had previously adapted to both HLA-A03 and B57 immune responses (Figure 9A; Table S8 in Text S1). Reversion of other well-described escape mutations such as I₂₉₃T in the Pol B51-TI8 epitope [41,42] and $I_{63}T$ in the Vpr A02-AL9 epitope [43] was also detected. In total, 15% (56/373) of all transmitted, non-consensus mutations exhibited evolution consistent with reversion over the four years of follow-up (Figure 9B). Thus, the increased sensitivity afforded by the longitudinal deep sequencing data revealed that not only is reversion of transmitted mutations a significant contributor to the evolution of HIV-1, but that these mutations revert at vastly different rates implying significantly different impacts of each mutation on viral replication capacity.

As a result of these findings we undertook a closer examination of the evolution within the ten targeted CD8 epitopes. Transmitted mutations at baseline were in fact present in five of these epitopes (Table S6 in Text S1), with evolution in two of the chronically targeted epitopes consistent with the reversion of transmitted mutations. In the Env A01-RY9 epitope, despite the fact that CD8+ T cell responses were not detected until day 476, as early as day 59 low frequency mutations developed at the residues containing transmitted mutations (Figure 7, Table S6 in Text **S1**). In line with the hypothesis that the early evolution in this epitope may have represented reversion of transmitted mutations, we first detected low magnitude (70 SFC) immune responses against this Env A01-RY9 epitope at day 476 following partial outgrowth of the HIV-1B consensus residue (R_{794} at 9%; **Tables** S5 and S6 in Text S1). Similarly, in the other late-targeted Env A24-RL9 epitope we observed partial reversion (20%) towards consensus of another transmitted mutation (K₅₉₃R) at day 476,

which was also associated with the late development of a low magnitude (84 SFC) response against the wild-type form of the epitope (**Table S5 and S6** in **Text S1**). Thus, while the transmission of mutations within some CD8 epitopes restricted by subject 9213 prevented the mounting of early immune responses to these epitopes, the reversion of transmitted mutations, even at very low frequencies, was sufficient to enable the priming of immune responses to these epitopes.

Discussion

We have established a high-throughput deep sequencing platform to assess HIV-1 sequence diversity across the entire HIV-1 genome. As the result of developing novel sequence assembly and variant detection algorithms, we were able to rapidly produce deep sequence data for a diverse set of 89 clade B clinical isolates and to dissect the evolutionary dynamics of HIV-1 during the earliest stages of acute infection. Our results from an in-depth analysis of a single subject reveal that the majority of early, low frequency mutations arising during the acute phase of infection reflect adaptation to host CD8+ T cell responses. Moreover, the temporal link observed between interruption of the decline in peak viremia and escape from the most immunodominant CD8+ T cell responses through low-frequency mutations suggests that the rate of escape from a few key acute phase CD8+ T cell responses may strongly influence primary control of HIV-1, and potentially viral set point. Thus, immune control during acute HIV-1 infection may be substantially influenced by early viral adaptations not detected by conventional sequencing approaches.

The central role of cellular immune responses in the early control of HIV-1 is highlighted by our findings that across the viral proteome the majority of early, low frequency adaptive mutations in subject 9213 were associated with CD8+ T cell responses. These data support the substantial selective pressure exerted upon HIV-1 by these responses early after infection. While limited sample availability precluded an analysis of CD4+ T cell responses, none of the rapidly evolving sites in subject 9213 arose exclusively within described CD4 T cell epitopes. While this does not exclude the possibility of CD4 escape, our data were not able to directly identify any evidence of CD4 escape. Recent studies have illustrated that HIV-specific CD8+ T cell responses are guided by distinct immunodominance hierarchies, whereby certain responses consistently arise more rapidly during the acute phase of infection, and can even dominate responses restricted by other HLA alleles [44,45]. The two most immunodominant described B38- and A24restricted epitopes, Vif B38-WI9 and Nef A24-RW8, were also found to be immunodominant in subject 9213. Moreover, they also represented the most rapidly escaping epitopes, with the kinetics of viral escape in subject 9213 corresponding in general to the hierarchy of all CD8+ T cell responses at baseline. These data support a strong link between the strength of a response and the relative selection pressure exerted, in line with recent data by Ferrari et al. [46]. More importantly, the observation that cessation of the rapid decline from peak viremia in subject 9213 was temporally coincident with viral escape from these two most immunodominant CD8+ T cell responses suggests that the duration of effectiveness of such immunodominant responses may be critical to the successful containment of early viral replication and prolonged viral load decline. Thus, the rate at which the earliest immunodominant CD8+ T cell responses are lost through viral escape may substantially influence the establishment of viral load set point, and thus progression to AIDS [47]. Unfortunately, with the exception of a few protective HLA alleles, the majority of immunodominant CD8 epitopes occur within more variable regions of the virus that would be expected to escape rapidly because they impart little or no viral fitness cost. As such, our data revealing that combinations of low frequency adaptive amino acid mutations may critically impact early control of HIV-1 by subverting the key CD8+ T cell responses may help to explain the inability of most HLA alleles to fully suppress early viral replication.

Characterization of the molecular pathways of viral escape is central to the rational design of a durable T-cell based vaccine. The sensitivity of our approach revealed a common pattern of evolution within the majority of escaping epitopes, including both immunodominant and subdominant responses, in which combinations of multiple low frequency escape mutations were replaced over time by HLA-anchor mutations. CD8+ T cell responses specific for the earlier escape variants were associated with selection of these "secondary" escape mutations that were substantially more effective in abrogating CTL recognition. These data, and prior reports of variant-specific responses [37-40], reveal the efficacy of these variant-specific responses, and suggest a potentially more important role for these responses in the control of HIV-1. It is important to note, however, that while some studies have carefully demonstrated the ability of the immune response to recognize CTL escape variants using tetramers and peptide dilutions [37–39], other studies have found that the high peptide concentrations often used to detect cross-reactive responses to variants can be misleading since the peptide levels are often substantially higher than physiological levels [48,49]. Unfortunately, a lack of sample availability at the early time points when these responses were robust prevented the testing of mutant and autologous peptides at additional dilutions. Therefore, it will be important in future studies to examine the recognition of these types of early CTL escape mutations using physiological peptide concentrations of peptides, or point-mutant strains of HIV-1 so that the mutant epitopes can be naturally processed and presented at the cell surface at physiological levels. Nonetheless, these data exemplify the continuous nature of host-virus co-adaptation and suggest the need to consider these early transient escape mutations when designing vaccine immunogens. For example, mosaic immunogen approaches [50,51], designed to impede viral escape by inducing responses against early escape mutations, may benefit from inclusion of these transient low frequency variants that are likely absent from the larger chronic sequence datasets upon which mosaic vaccine antigens are based. Similarly, these deep sequence data provide greater insight into the critical role of compensatory mutations, whereby viral escape within structurally interacting regions of a protein requires one or more co-evolving secondary mutations to retain protein structure and function [10,52,53]. In the Nef A24-RW8 epitope, eventual development of the position 2 HLA-anchor mutation (Y144F) was tightly linked to an upstream $I_{142}T$ mutation, exclusively present on the haplotype expressing the escape mutant (Figure 6D). Thus, supplementing existing HIV-1 sequence databases with deep sequence data from both acute and chronically infected individuals may help to identify regions of HIV-1 which require co-evolving sites to escape [54] and thus would be most susceptible to immune targeting [55,56].

Transmitted escape mutations can also influence the course of infection both by impairing the induction of CD8+ T cell responses [35,43,57], but also by attenuating viral replication capacity [58,59]. Importantly, the rate at which transmitted mutations revert may serve as a more accurate in vivo measurement of the relative impact of these mutations on viral fitness, as compared to in vitro viral fitness measurements [10]. The range of reversion rates of transmitted mutations observed in this genomewide study $(0.0887 \text{ to } 0.0015 \text{ day}^{-1})$, including some that were very rapid, supports a significant impact of some of these mutations on viral replication capacity. The ability to more accurately determine the true rates of genome-wide reversions using the more sensitive deep sequencing data provides the unique opportunity to now systematically quantify the contribution of transmitted mutations on viral fitness, which may provide additional insight into the potentially significant contribution of viral genotype to HIV-1 set-point viral load [60].

The deep sequencing approach presented here yields results consistent with those of traditional cloning or SGA. A recent study by Jordan et al illustrates similar results for sequence diversity detection between standard PCR/cloning and SGA [61]. While improving upon the sensitivity of these methods, and providing the ability to simultaneously assay genetic diversity across all residues in the genome, our variant detection methods achieve a sensitivity and specificity of >97% at a substantially reduced cost as compared to SGA or cloning. Despite this high accuracy, as with other sequencing approaches, deep sequencing has its own set of limitations. First, despite efforts to optimize read alignments, misalignments can occur especially at the ends of amplicons and reads and lead to false positives; V-phaser is designed to limit false positives and Macalalad et al. (manuscript submitted) have shown that the variant detection methods described here achieve a positive predicted value (PPV) of 98%. Second, 454 deep sequencing is constrained in its ability to identify long-range linked mutations beyond a single read length of approximately 400 bp. When compared to SGA, this may limit its utility to understand more complex haplotype interactions, such as whether escape mutations in two simultaneously escaping eptiopes are arising upon the same viral haplotype [7], or upon distinct viral haplotypes which later recombine [38]. Here, deep sequencing approaches and SGA may well serve to complement their respective individual strengths. Third, the bulk amplified PCR products used for this deep sequencing approach may be more prone to in vitro recombination events than the single-template amplifications used during SGA [62]. While this is unlikely to alter

the frequency of variants detected by deep sequencing, it could limit the ability to accurately assess in vivo recombination rates and longer viral haplotypes. However, since both bulk amplification and SGA approaches rely on the bulk reverse transcription (RT) of RNA to cDNA, which itself may be prone to *in vitro* recombination [63], both deep sequencing and SGA approaches may still be susceptible to recombination events. Finally, given the ability to routinely sequence the viral quasispecies at near unlimited depth the issue of template resampling may be a concern. In subject 9213 we quantified the number of input RNA template molecules used for each cDNA synthesis. In each case the number of template molecules (>1000 RNA copies) was greater than the fold depth of sequence data achieved (535±325 reads), arguing against template resampling having unduly influenced our findings. Supporting this conclusion is the congruence in variants and variant frequencies observed across 454, clonal, and SGA data sets (see Figure 2, and Supplementary Results in **Text S1**).

The accuracy of the deep sequencing methods described here to identify variable and conserved sites are further confirmed by comparison of the diversity detected within individual patients to that observed in the global HIV-1 population. As shown in Figure S2 in Text S1, which illustrates diversity plots for the 89 clade B clinical isolates, consistent diversity "hotspots" were observed in each protein, including the 5' (p17) and 3' (p15) regions of Gag and the V1-V3 loops of Env. Notably, sites that frequently exhibited high intra-patient diversity were more likely to be highly polymorphic in consensus sequences of circulating strains when compared both across the whole genome (Wilcoxon, p < 0.0001) and within any gene (Wilcoxon, p < 0.01). Conversely, 28 residues were entirely conserved in both the intra-patient and global datasets. Such data support the accuracy of the deep sequencing methods and also provide a comprehensive view of the extent of genome-wide intra-host sequence diversity achieved during chronic HIV-1 infection, revealing that sites commonly susceptible to intra-host diversity contribute directly to the diversity observed between circulating strains.

The development of a robust genome-wide HIV-1 deep sequencing approach provides both the means to rapidly produce whole genome data for large cohorts and a unique opportunity to sensitively and globally profile HIV-1's earliest adaptations to host immune pressures. Genome-wide diversity profiles may serve as a sensitive and effective readout of host immunity during both natural infection, but also following vaccination such as in the case of breakthrough subjects from the HIV-1 STEP trial [64]. Our analysis of early sequence evolution in a single subject indicates that a small number of early specific CD8+ T cell responses represent the major selective force being evaded when peak HIV-1 viremia first comes under control. Extending these results to larger cohorts of individuals, especially in subjects naturally controlling HIV-1 following acute infection, would support a critical role for the

References

- Allen TM, O'Connor DH, Jing P, Dzuris JL, Mothe BR, et al. (2000) Tatspecific cytotoxic T lymphocytes select for SIV escape variants during resolution of primary viraemia. Nature 407: 386–390.
- Goulder PJ, Phillips RE, Colbert RA, McAdam S, Ogg G, et al. (1997) Late escape from an immunodominant cytotoxic T-lymphocyte response associated with progression to AIDS. Nat Med 3: 212–217.
- Timm J, Lauer GM, Kavanagh DG, Sheridan I, Kim AY, et al. (2004) CD8 epitope escape and reversion in acute HCV infection. J Exp Med 200: 1593–1604.
- Kwiatkowski D, Marsh K (1997) Development of a malaria vaccine. Lancet 350: 1696–1701.
- Allen TM, Altfeld M, Geer SC, Kalife ET, Moore C, et al. (2005) Selective escape from CD8+ T-cell responses represents a major driving force of human immunodeficiency virus type 1 (HIV-1) sequence diversity and reveals constraints on HIV-1 evolution. J Virol 79: 13239–13249.

maintenance of a few key CD8+ T cell responses in the critical control of HIV-1. If so, vaccine strategies aimed at triggering immunodominant responses against critical regions of the virus may prove more effective than efforts attempting to maximize the breadth or polyfunctionality of vaccine-elicited CD8+ T cell responses [55].

Materials and Methods

Ethics statement

All subjects gave written informed consent and the study was approved by the Massachusetts General Hospital Review Board.

Study subjects

Plasma samples were obtained from HIV-1 cohorts at the Massachusetts General Hospital in Boston, Massachusetts, the Jessen-Praxis in Berlin, Germany and the HIV Swiss Cohort. Subject 9213 was identified during primary HIV-1 infection (Western Blot negative; Fiebig II–III) [29], and time points are defined from day of presentation with symptomatic acute HIV-1 infection. High and intermediate-resolution HLA class I genotyping was performed by sequence-specific PCR and direct sequencing according to standard procedures.

See **Text S1** for a detailed description of sample preparation, library construction, and sequencing protocols, as well as a description of the genome assembly and variant detection algorithms and their validation.

Supporting Information

Text S1 Supplementary document containing detailed materials and methods as well as supplementary results, tables and figures. (DOC)

Acknowledgments

We would like to thank the International HIV Controllers Study and the Swiss HIV Cohort Study (SHCS) for collaboration, the Broad Genome Sequencing and Biological Sample Platforms for providing sample handling and sequencing support, Eunice Pae and Ana Maldonado from the Fenway Community Health Care Center for referral of study subjects, and Leslie Gaffney for assistance with preparation of the figures.

Author Contributions

Conceived and designed the experiments: TMA MRH CLB PC NJL KAP ARM MCZ. Performed the experiments: CMM EMR RLE LMG AB YW MC HS AKB TD RN ADG LB MK. Analyzed the data: TMA MRH CLB PC NJL KAP ARM AMB CMM EMR SG MCZ RLE YW HS DT RN QZ TPS SG JZL SKY MA. Contributed reagents/materials/analysis tools: KLA CZ OG CB CH HFG ZLB CJB SB JR JPT KHM ER FP HJ BWB BDW. Wrote the paper: TMA MRH CLB PC NJL KAP ARM MCZ BWB BDW.

- Price DA, Goulder PJ, Klenerman P, Sewell AK, Easterbrook PJ, et al. (1997) Positive selection of HIV-1 cytotoxic T lymphocyte escape variants during primary infection. Proc Natl Acad Sci U S A 94: 1890–1895.
- Goonetilleke N, Liu MK, Salazar-Gonzalez JF, Ferrari G, Giorgi E, et al. (2009) The first T cell response to transmitted/founder virus contributes to the control of acute viremia in HIV-1 infection. J Exp Med 206: 1253–1272.
- Borrow P, Lewicki H, Wei XP, Horwitz MS, Peffer N, et al. (1997) Antiviral pressure exerted By HIV-1-specific cytotoxic T lymphocytes (CTLs) during primary infection demonstrated by rapid selection of CTL escape virus. Nat Med 3: 205–211.
- Richman DD, Wrin T, Little SJ, Petropoulos CJ (2003) Rapid evolution of the neutralizing antibody response to HIV type 1 infection. Proc Natl Acad Sci U S A 100: 4144–4149.
- Schneidewind A, Brockman MA, Yang R, Adam RI, Li B, et al. (2007) Escape from the dominant HLA-B27-restricted cytotoxic T-lymphocyte response in

- Martinez-Picado J, Prado JG, Fry EE, Pfafferott K, Leslie A, et al. (2006) Fitness cost of escape mutations in p24 Gag in association with control of human immunodeficiency virus type 1. J Virol 80: 3617–3623.
- Troyer RM, McNevin J, Liu Y, Zhang SC, Krizan RW, et al. (2009) Variable fitness impact of HIV-1 escape mutations to cytotoxic T lymphocyte (CTL) response. PLoS Pathog 5: e1000365.
- Miura T, Brockman MA, Schneidewind A, Lobritz M, Pereyra F, et al. (2009) HLA-B57/B*5801 human immunodeficiency virus type 1 elite controllers select for rare gag variants associated with reduced viral replication capacity and strong cytotoxic T-lymphocyte [corrected] recognition. J Virol 83: 2743–2755.
- Liu Y, McNevin J, Cao J, Zhao H, Genowati I, et al. (2006) Selection on the human immunodeficiency virus type 1 proteome following primary infection. J Virol 80: 9519–9529.
- Palmer S, Kcarney M, Maldarelli F, Halvas EK, Bixby CJ, et al. (2005) Multiple, linked human immunodeficiency virus type 1 drug resistance mutations in treatment-experienced patients are missed by standard genotype analysis. J Clin Microbiol 43: 406–413.
- Salazar-Gonzalez JF, Bailes E, Pham KT, Salazar MG, Guffey MB, et al. (2008) Deciphering human immunodeficiency virus type 1 transmission and early envelope diversification by single-genome amplification and sequencing. J Virol 82: 3952–3970.
- Margulies M, Egholm M, Altman WE, Attiya S, Bader JS, et al. (2005) Genome sequencing in microfabricated high-density picolitre reactors. Nature 437: 376–380.
- Simen BB, Simons JF, Hullsiek KH, Novak RM, Macarthur RD, et al. (2009) Low-abundance drug-resistant viral variants in chronically HIV-infected, antiretroviral treatment-naive patients significantly impact treatment outcomes. J Infect Dis 199: 693–701.
- Poon AF, Swenson LC, Dong WW, Deng W, Kosakovsky Pond SL, et al. (2010) Phylogenetic analysis of population-based and deep sequencing data to identify coevolving sites in the nef gene of HIV-1. Mol Biol Evol 27: 819–832.
- Wang C, Mitsuya Y, Gharizadeh B, Ronaghi M, Shafer RW (2007) Characterization of mutation spectra with ultra-deep pyrosequencing: application to HIV-1 drug resistance. Genome Res 17: 1195–1201.
- Hedskog C, Mild M, Jernberg J, Sherwood E, Bratt G, et al. (2010) Dynamics of HIV-1 quasispecies during antiviral treatment dissected using ultra-deep pyrosequencing. PLoS One 5: e11345.
- Tsibris AM, Korber B, Arnaout R, Russ C, Lo CC, et al. (2009) Quantitative deep sequencing reveals dynamic HIV-1 escape and large population shifts during CCR5 antagonist therapy in vivo. PLoS One 4: e5683.
- Bimber BN, Burwitz BJ, O'Connor S, Detmer A, Gostick E, et al. (2009) Ultradeep pyrosequencing detects complex patterns of CD8+ T-lymphocyte escape in simian immunodeficiency virus-infected macaques. J Virol 83: 8247–8253.
- Love TM, Thurston SW, Keefer MC, Dewhurst S, Lee HY (2010) Mathematical Modeling of Ultradeep Sequencing Data Reveals That Acute CD8+ T-Lymphocyte Responses Exert Strong Selective Pressure in Simian Immunodeficiency Virus-Infected Macaques, but Still Fail to Clear Founder Epitope Sequences. J Virol 84: 5802–5814.
- Bimber BN, Dudley DM, Lauck M, Becker EA, Chin EN, et al. (2010) Whole genome characterization of HIV/SIV intra-host diversity by ultra-deep pyrosequencing. J Virol 84: 12087–92.
- Fischer W, Ganusov VV, Giorgi EE, Hraber PT, Keele BF, et al. (2010) Transmission of single HIV-1 genomes and dynamics of early immune escape revealed by ultra-deep sequencing. PLoS One 5: e12303.
- Mitsuya Y, Varghese V, Wang C, Liu TF, Holmes SP, et al. (2008) Minority human immunodeficiency virus type 1 variants in antiretroviral-naive persons with reverse transcriptase codon 215 revertant mutations. J Virol 82: 10747–10755.
- Altshuler D, Pollara VJ, Cowles CR, Van Etten WJ, Baldwin J, et al. (2000) An SNP map of the human genome generated by reduced representation shotgun sequencing. Nature 407: 513–516.
- Fiebig EW, Wright DJ, Rawal BD, Garrett PE, Schumacher RT, et al. (2003) Dynamics of HIV viremia and antibody seroconversion in plasma donors: implications for diagnosis and staging of primary HIV infection. Aids 17: 1871–1879.
- Keele BF, Giorgi EE, Salazar-Gonzalez JF, Decker JM, Pham KT, et al. (2008) Identification and characterization of transmitted and early founder virus envelopes in primary HIV-1 infection. Proc Natl Acad Sci U S A 105: 7552–7557.
- Gottlieb GS, Heath L, Nickle DC, Wong KG, Leach SE, et al. (2008) HIV-1 variation before seroconversion in men who have sex with men: analysis of acute/early HIV infection in the multicenter AIDS cohort study. J Infect Dis 197: 1011–1015.
- 32. Abrahams MR, Anderson JA, Giorgi EE, Scoighe C, Mlisana K, et al. (2009) Quantitating the multiplicity of infection with human immunodeficiency virus type 1 subtype C reveals a non-poisson distribution of transmitted variants. J Virol 83: 3556–3567.
- Haaland RE, Hawkins PA, Salazar-Gonzalez J, Johnson A, Tichacek A, et al. (2009) Inflammatory genital infections mitigate a severe genetic bottleneck in heterosexual transmission of subtype A and C HIV-1. PLoS Pathog 5: e1000274.

- Liu Y, McNevin JP, Holte S, McElrath MJ, Mullins JI (2011) Dynamics of viral evolution and CTL responses in HIV-1 infection. PLoS One 6: e15639.
- Allen TM, Altfeld M, Yu XG, O'Sullivan KM, Lichterfeld M, et al. (2004) Selection, transmission, and reversion of an antigen-processing cytotoxic Tlymphocyte escape mutation in human immunodeficiency virus type 1 infection. J Virol 78: 7069–7078.
- Zimbwa P, Milicic A, Frater J, Scriba TJ, Willis A, et al. (2006) Precise Identification Of An Hiv-1 Antigen Processing Mutant. J Virol 81: 2031–2038.
- Allen TM, Yu XG, Kalife ET, Reyor LL, Lichterfeld M, et al. (2005) De novo generation of escape variant-specific CD8+ T-cell responses following cytotoxic T-lymphocyte escape in chronic human immunodeficiency virus type 1 infection. J Virol 79: 12952–12960.
- Streeck H, Li B, Poon AF, Schneidewind A, Gladden AD, et al. (2008) Immunedriven recombination and loss of control after HIV superinfection. J Exp Med 205: 1789–1796.
- Feeney ME, Tang Y, Pfafferott K, Roosevelt KA, Draenert R, et al. (2005) HIVl viral escape in infancy followed by emergence of a variant-specific CTL response. J Immunol 174: 7524–7530.
- Turnbull EL, Lopes AR, Jones NA, Cornforth D, Newton P, et al. (2006) HIV-1 epitope-specific CD8+ T cell responses strongly associated with delayed disease progression cross-recognize epitope variants efficiently. J Immunol 176: 6130–6146.
- Brumme ZL, John M, Carlson JM, Brumme CJ, Chan D, et al. (2009) HLAassociated immune escape pathways in HIV-1 subtype B Gag, Pol and Nef proteins. PLoS One 4: e6687.
- Kawashima Y, Pfafferott K, Frater J, Matthews P, Payne R, et al. (2009) Adaptation of HIV-1 to human leukocyte antigen class I. Nature 458: 641–645.
- 43. Altfeld M, Allen TM, Kalife ET, Frahm N, Addo MM, et al. (2005) The majority of currently circulating human immunodeficiency virus type 1 clade B viruses fail to prime cytotoxic T-lymphocyte responses against an otherwise immunodominant HLA-A2-restricted epitope: implications for vaccine design. J Virol 79: 5000–5005.
- 44. Streeck H, Jolin JS, Qi Y, Yassine-Diab B, Johnson RC, et al. (2009) Human immunodeficiency virus type 1-specific CD8+ T-cell responses during primary infection are major determinants of the viral set point and loss of CD4+ T cells. J Virol 83: 7641–7648.
- Altfeld M, Kalife ET, Qi Y, Streeck H, Lichterfeld M, et al. (2006) HLA Alleles Associated with Delayed Progression to AIDS Contribute Strongly to the Initial CD8(+) T Cell Response against HIV-1. PLoS Med 3: e403.
- 46. Ferrari G, Korber B, Goonetilleke N, Liu MK, Turnbull EL, et al. (2011) Relationship between functional profile of HIV-1 specific CD8 T cells and epitope variability with the selection of escape mutants in acute HIV-1 infection. PLoS Pathog 7: e1001273.
- Mellors J, Rinaldo C, Gupta P, White R, Todd mJ, et al. (1996) Prognosis in HIV-1 infection predicted by the quantity of virus in plasma. Science 272: 1167–1170.
- Friedrich TC, McDermott AB, Reynolds MR, Piaskowski S, Fuenger S, et al. (2004) Consequences of cytotoxic T-lymphocyte escape: common escape mutations in simian immunodeficiency virus are poorly recognized in naive hosts. J Virol 78: 10064–10073.
- Bennett MS, Ng HL, Ali A, Yang OO (2008) Cross-clade detection of HIV-1specific cytotoxic T lymphocytes does not reflect cross-clade antiviral activity. J Infect Dis 197: 390–397.
- Santra S, Liao HX, Zhang R, Muldoon M, Watson S, et al. (2010) Mosaic vaccines elicit CD8+ T lymphocyte responses that confer enhanced immune coverage of diverse HIV strains in monkeys. Nat Med 16: 324–328.
- Fischer W, Perkins S, Theiler J, Bhattacharya T, Yusim K, et al. (2007) Polyvalent vaccines for optimal coverage of potential T-cell epitopes in global HIV-1 variants. Nat Med 13: 100–106.
- Brockman MA, Schneidewind A, Lahaie M, Schmidt A, Miura T, et al. (2007) Escape and compensation from early HLA-B57-mediated cytotoxic Tlymphocyte pressure on human immunodeficiency virus type 1 Gag alter capsid interactions with cyclophilin A. J Virol 81: 12608–12618.
- Carlson JM, Brumme ZL, Rousseau CM, Brumme CJ, Matthews P, et al. (2008) Phylogenetic dependency networks: inferring patterns of CTL escape and codon covariation in HIV-1 Gag. PLoS Comput Biol 4: e1000225.
- 54. Schneidewind A, Brockman MA, Sidney J, Wang YE, Chen H, et al. (2008) Structural and functional constraints limit options for cytotoxic T-lymphocyte escape in the immunodominant HLA-B27-restricted epitope in human immunodeficiency virus type 1 capsid. J Virol 82: 5594–5605.
- Altfeld M, Allen TM (2006) Hitting HIV where it hurts: an alternative approach to HIV vaccine design. Trends Immunol 27: 504–510.
- Dahirel V, Shekhar K, Pereyra F, Miura T, Artyomov M, et al. (2011) From the Cover: Coordinate linkage of HIV evolution reveals regions of immunological vulnerability. Proc Natl Acad Sci U S A 108: 11530–11535.
- Leslie A, Kavanagh D, Honeyborne I, Pfafferott K, Edwards C, et al. (2005) Transmission and accumulation of CTL escape variants drive negative associations between HIV polymorphisms and HLA. J Exp Med 201: 891–902.
- Goepfert PA, Lumm W, Farmer P, Matthews P, Prendergast A, et al. (2008) Transmission of HIV-1 Gag immune escape mutations is associated with reduced viral load in linked recipients. J Exp Med 205: 1009–1017.
- Chopera DR, Woodman Z, Milsana K, Mlotshwa M, Martin DP, et al. (2008) Transmission of HIV-1 CTL escape variants provides HLA mis-matched recipients with a survival advantage. PLoS Pathog 4: e1000033.

- Alizon S, von Wyl V, Stadler T, Kouyos RD, Yerly S, et al. (2010) Phylogenetic approach reveals that virus genotype largely determines HIV set-point viral load. PLoS Pathog 6: e1001123.
- Jordan MR, Kearney M, Palmer S, Shao W, Maldarelli F, et al. (2010) Comparison of standard PCR/cloning to single genome sequencing for analysis of HIV-1 populations. J Virol Methods 168: 114–120.
- Smyth RP, Schlub TE, Grimm A, Venturi V, Chopra A, et al. (2010) Reducing chimera formation during PCR amplification to ensure accurate genotyping. Gene 469: 45–51.
- Negroni M, Ricchetti M, Nouvel P, Buc H (1995) Homologous recombination promoted by reverse transcriptase during copying of two distinct RNA templates. Proc Natl Acad Sci U S A 92: 6971–6975.
- Rolland M, Tovanabutra S, deCamp AC, Frahm N, Gilbert PB, et al. (2011) Genetic impact of vaccination on breakthrough HIV-1 sequences from the STEP trial. Nat Med 17: 366–371.
- Asquith B, Edwards CT, Lipsitch M, McLean AR (2006) Inefficient cytotoxic T lymphocyte-mediated killing of HIV-1-infected cells in vivo. PLoS Biol 4: e90.